Author's response to reviews

Title: Use of dexmedetomidine for prophylactic analgesia and sedation in delayed extubation patients after craniotomy: a study protocol and statistical analysis plan for a randomized controlled trial

Authors:

Li-Hong Zhao (zhaolh.cn@gmail.com)
Zhong-Hua Shi (katelynszh@163.com)
Ning-Ning Yin (yinningning2008@126.com)
Jian-Xin Zhou (zhoujx.cn@gmail.com)

Version: 2 Date: 26 July 2013

Author's response to reviews:

Dear Editor:

Responses to reviewer’s comments

Minor Essential Revisions:

The distribution of the data (especially the primary outcome variable) should be tested for normality before parametric statistical tests are employed.

Response:

Have been revised in manuscript, and marked with yellow highlight.

Additionally, some grammatical errors have been revised and marked with yellow highlight.